Author’s response to reviews

**Title:** Human adipose-derived mesenchymal stem cells attenuate collagen antibody-induced autoimmune arthritis by inducing expression of FCGIIB receptors

**Authors:**

- Hyoju Yi (yallashung@gmail.com)
- Kwi Young Kang (kykang@catholic.ac.kr)
- Hyerin Jung (ilovehyelin@gmail.com)
- Yeri Alice Rim (llyerill0114@gmail.com)
- Narae Park (narae0801@naver.com)
- Juryun Kim (juryunkim1@gmail.com)
- Seung Min Jung (pobi28@catholic.ac.kr)
- Sung-Hwan Park (rapark@catholic.ac.kr)
- Youngkyun Kim (starbio98@gmail.com)
- Ji Hyeon Ju (juji@catholic.ac.kr)

**Version:** 6  
**Date:** 11 July 2015

**Author’s response to reviews:** see over
1) Authors' contributions:

(i) in this section, the sentence 'MT participated in the sequence alignment.' is included but MT does not appear to be one of the authors on the paper. Please advise.

Answer: We made a mistake. We corrected the mistake. Thank you.

(ii) please could you ensure that the ordering of the initials in this section is in the correct format [author's initial of first name(s) and then initial of surname; i.e. HY, KYK etc.]

Answer: We changed the ordering of the initials according to the guideline.

2) Abstract: please change 'Introduction' to 'Background'

Answer: We changed the 'Introduction' to 'Background' in Abstract section

3) Main text: please change 'Introduction' to 'Background'

Answer: We changed the 'Introduction' to 'Background' in Main text.